Author's response to reviews

Title: Genetic Risk Factors for Cerebrovascular Disease in Children with Sickle Cell Disease: Design of a Case-Control Association Study and Genomewide Screen

Authors:

Gaye T Adams (gayetadams@hotmail.com)
Dr Harold Snieder (hsnieder@mcg.edu)
Virgil C McKie (vmckie@mcg.edu)
Betsy A Clair (bclair@mcg.edu)
Donald Brambilla (donb@neri.org)
Robert J Adams (rjadams@mcg.edu)
Ferdane Kutlar (futlar@mcg.edu)
Dr Abdullah Kutlar (akutlar@mcg.edu)

Version: 4 Date: 3 Jul 2003

Dear Editor,

Thank you for your e-mail of 7/3/03 informing me that our manuscript has been accepted. The requested revisions, moving of The List of Abbreviations and adding a section about the additional files, has been completed.

Please let me know if there are any other changes necessary.

Sincerely,
Abdullah Kutlar